Author's response to reviews

Title: Loss of international medical experiences: knowledge, skills, and attitudes at risk

Authors:

Corita R. Grudzen (cgrudzen@mednet.ucla.edu)
Eric Legome (legome01@med.nyu.edu)

Version: 6 Date: 26 November 2007

Author's response to reviews:

The references were changed in the text so that they are cited using consecutive numbers in square brackets. Large groups of references were reformatted as requested.